Author’s response to reviews

Title: Fibrinogen Caracas IX. A novel fibrinogen mutation: FGA g. 3057 C>T (p. Arg104>Cys) impair fibrinogen secretion

Authors:

Rita Marchi (marchi.rita@gmail.com)

Maryuri Linares (maryuri.linares.26@gmail.com)

Héctor Rojas (hectorrafaelrojas@gmail.com)

Arlette Ruiz-Sáez (arletteruizsaez@gmail.com)

Michael Meyer (mica.mey@web.de)

Alessandro Casini (alessandro.casini@hcuge.ch)

Stephen Brennan (steve.brennan@otago.ac.nz)

Version: 2 Date: 26 Jul 2017

Author’s response to reviews:

Editor:

1. Supplementary videos

Thank you for providing your mp4 video files as part of the manuscript supplementary information. Unfortunately, we are unable to open and view these on our system - please ensure that these videos are fully functioning before being uploaded.

We accept movies in mp4, mpeg, mov, avi, swf and animated GIF formats. MOV is a common multimedia format, which is often used for saving movies and other video files. This format uses a proprietary compression algorithm developed by Apple Computer; compatible with both Macintosh and Windows platforms. MPG is a common digital video format, which typically incorporates MPEG-1 or MPEG-2 audio and video compression. AVI files can contain both audio and video data in a file container that allows synchronous audio-with-video playback.

Answer:

We have included the movies in mp4 format. We apologise for the inconvenient.
Reviewer reports:

Eduardo Angles-Cano, M.D., Sc.D. (Reviewer 1): The authors have satisfactorily responded to my critics and comments.

I have no further questions.

Johnny Mahlangu (Reviewer 2): Please include all comments for the authors in this box rather than uploading your report as an attachment. Please only upload as attachments annotated versions of manuscripts, graphs, supporting materials or other aspects of your report which cannot be included in a text format. Please overwrite this text when adding your comments to the authors.